Author’s response to reviews

Title: Diffuse large B-cell lymphoma of the colon with synchronous liver metastasis: a rare case report mimicking metastatic colorectal adenocarcinoma.

Authors:

Domenico Risio (domenicorisio@virgilio.it)
Rossana Percario (domenicorisio@virgilio.it)
Margherita Legnini (domenicorisio@virgilio.it)
Francesco Caldaralo (domenicorisio@virgilio.it)
Domenico Angelucci (domenicorisio@virgilio.it)
Camilla Marinelli (domenicorisio@virgilio.it)
Alberto D'Aulerio (domenicorisio@virgilio.it)
Roberto Cotellese (domenicorisio@virgilio.it)
Luca Napolitano (domenicorisio@virgilio.it)
Paolo Innocenti (innocenti@unich.it)

Version: 4
Date: 11 September 2014

Author’s response to reviews: see over
September 8th, 2014

Dear Editor,

please find here attached an updated version of the manuscript (MS: 1075262135133169) “Diffuse large B-cell lymphoma of the colon with synchronous liver metastasis: a rare case report mimicking metastatic colorectal adenocarcinoma” revised according to your recommendations.

As suggested, the document has been also copyedited by a native English speaking colleague, who is mentioned in the acknowledgements. Sciaudone’s manuscript has been added in the discussion, highlighting the importance of considering lymphomatous polyposis in differential diagnosis of patients affected by atypical inflammatory bowel disease.

So, the major changes are as follows.

1. we have removed consent form
2. we have changed the format of the references according to the sample.
3. we have corrected some scattered spelling and grammatical errors
4. we have reported in the text the paper by Sciaudone G et al.
5. Acknowledgments has been introduced.

We believe that our manuscript, describing a rare case of colonic lymphoma with synchronous liver metastasis, highlights the difficulty in diagnosing of primary colonic lymphoma. We think that the most important point of this case report is to underline that this kind of tumour should be considered in the differential diagnosis of colon mass with with enlarged lymph nodes in the abdomen or pelvis and synchronous liver nodules when
the colonoscopy biopsies were not diagnostic. We believe this paper may be of particular interest to the readers of your journal.

The manuscript is not being reviewed or considered for publication by any other journal, and has not been published elsewhere. The manuscript has been read and approved by all the authors, who declare no conflict of interest.

Thank you very much for your notable cooperation and support that you have provided.

We are satisfied for publication on your Journal.

Best Regards

Paolo Innocenti, MD
Surgical Sciences Department
University “G. d’Annunzio”, Chieti
Via dei Vestini 13
66100 CHIETI (Italy)
Phone: +39-0871-358062
Fax: +39-0871-572126
E-mail: innocenti@unich.it